Author's response to reviews

Title: Delay of sputum smear conversion and outcomes of smears positive tuberculosis patients: A retrospective cohort study in Bafousam, Cameroon

Authors:

Fabrice N Djouma (nembotfabrice2009@yahoo.fr)
Michel Noubom (noubommichel@yahoo.fr)
Jérôme Ateudjieu (jateudi@yahoo.fr)
Hubert Donfack (donfackhubert@yahoo.fr)

Version: 2
Date: 5 August 2014

Author's response to reviews:

Dear Editor,

I am writing to submit the revised version of our manuscript entitled, “Delay of sputum smear conversion and outcomes of smears positive tuberculosis patients: A retrospective cohort study in Bafousam, Cameroon” for consideration for publication in BMC Infectious Diseases.

In this version of manuscript, following points are added:

• Ethical consideration
• Line numbering
• Acknowledgements chapter

Thank you for receiving our manuscript and considering it for review. We appreciate your time and look forward to your response.

Kind regards,

DJOUMA NEMBOT Fabrice, MPH
Department of Biomedical Sciences, University of Dschang
nembotfabrice2009@yahoo.fr